

PLEASE NOTE

This document, prepared by the <u>Legislative Counsel Office</u>, is an office consolidation of this regulation, current to February 1, 2004. It is intended for information and reference purposes only.

This document is *not* the official version of these regulations. The regulations and the amendments printed in the *Royal Gazette* should be consulted to determine the authoritative text of these regulations.

For more information concerning the history of these regulations, please see the *Table of Regulations*.

If you find any errors or omissions in this consolidation, please notify the Legislative Counsel Office at (902) 368-4291 or by email to <u>rmmacintyre@gov.p</u>e.ca

CHAPTER L-5

PRINCE EDWARD ISLAND LANDS PROTECTION ACT

FORMS REGULATIONS

Pursuant to clause 17(1)(c) of the *Prince Edward Island Lands Protection Act* R.S.P.E.I. 1988, Cap. L-5, Council made the following regulations:

1. The following forms are prescribed:

Forms

(a) Form 1 - application by or on behalf of a non-resident pursuant to section 4 of the *Prince Edward Island Lands Protection Act*;

(b) Form 2 - application by or on behalf of a corporation pursuant to section 5 of the *Prince Edward Island Lands Protection Act*;

(c) Form 3 - application for a permit to hold an aggregate land holding in excess of the limits set out in section 2 of the *Prince Edward Island Lands Protection Act*;

(d) Form 4A - land holding disclosure statement of a person pursuant to section 6.1 of the *Prince Edward Island Lands Protection Act*;

(e) Form 4B - land holding disclosure statement for a corporation pursuant to section 6.1 of the *Prince Edward Island Lands Protection Act*;

(f) Form 5A - self-calculating land holding disclosure statement for a person pursuant to subsection 11(3) of the *Prince Edward Island Lands Protection Act*;

(g) Form 5B - self-calculating land holding disclosure statement for a corporation pursuant to subsection 11(3) of the *Prince Edward Island Lands Protection Act*;

(h) Form 6A - land holding disclosure statement for a person pursuant to section 10 of the *Prince Edward Island Lands Protection Act* - short form;

(i) Form 6B - land holding disclosure statement for a corporation pursuant to section 10 of the *Prince Edward Island Lands Protection Act* - short form;

(j) Form 6C - land holding disclosure statement for a person pursuant to section 10 of the *Prince Edward Island Lands Protection Act*;

(k) Form 6D - land holding disclosure statement for a corporation pursuant to section 10 of the *Prince Edward Island Lands Protection Act*;

(l) Form 7A - self-calculating land holding disclosure statement for a person;

(m) Form 7B - self-calculating land holding disclosure statement for a corporation;

(n) Form 8 - application on behalf of a non-resident or corporation pursuant to clause 5.3(1)(b) of the *Lands Protection Act.* (EC219/96; 742/96;525/98)

Updated 2002

FORM 1

APPLICATION BY OR ON BEHALF OF A NON-RESIDENT PURSUANT TO SECTION 4 OF THE PRINCE EDWARD ISLAND Lands Protection Act

NON-RESIDENT APPLICANTS MAY BE REQUIRED TO ENTER INTO A LAND IDENTIFICATION AGREEMENT

NOTE: Multiple parcels may be included on one application form provided the transaction is between the same purchaser(s) and vendor(s). However, questions #9-18 should be answered for each parcel individually (attach a schedule if necessary).

 Citizenship:	ed through a trust in	dicate:				
7. Name of vendor (if different than present land holder):						
8. Address of vendor:						
		NT LAND HOLDER:				
		se ner (specify)				
(c) gift		ier (specify)				
11.Property Number		Shore frontage				
(from property tax bill)		e				
1		County				
		te the term of the lease: years				
		structures on each parcel:				
16. Indicate the present acreag	e breakdown of eac	h parcel separately:				
(a) cropland						
(b) pasture	(e) other					
(c) woodland						
17. State the intended use that						
		ommercial/industrial				
		abdivision of lots				
(c) agriculture						
18. Has each parcel been used	for agriculture in th	ne last five years?				

20. Indicate parcel number and acreage of each parcel of land in the Province in which the intended land holder and his minor children hold an interest, including land held by way of lease. The Act excludes any parcel that existed on May 1, 1995, of less than one acre that is located in a city or town. (Attach separate sheet if necessary.)

.....

21. For any corporation owning or leasing land within the Province in which the intended land holder or his minor children hold more than 5% of the issued voting shares, list on a separate sheet and attach:

(a) name and permanent address of the corporation;

(b) total number of voting shares issued by the corporation;

(c) total number of issued voting shares owned by the intended land holder and his minor children; and

parcel number and acreage of each parcel of land in the province now owned or leased by the corporation. The Act excludes any parcel that existed on May 1, 1995, of less than one acre that is located in a city or town.

22. Indicate any other circumstances that are relevant

23. I hereby certify that the above information and attached legal description of the property is true and correct in all respects.

Signature of applicant or his/her attorney

Date

Name of signatory (please print)

Attach (a) a legal description of the parcel to be acquired;

(b) an orthophoto showing the parcel to be acquired outlined in red;(c) a cheque made payable to the Island Regulatory and Appeals Commission in the amount of \$500.00 or 1% of the agreed purchase price, whichever is greater (refer

to Fees Regulations to determine if the fee is applicable).

RETURN COMPLETED FORM TO:

Island Regulatory and Appeals Commission

P. O. Box 577

Charlottetown, PE

C1A 7N8 Telephone: 902-892-3501

NOTE: In accordance with section 9.(1) of the P.E.I. Lands Protection Act special conditions may be required. section 9.(1) states:

9.(1) The Lieutenant Governor in Council may impose such conditions on a permit issued under section 4 or 5 as he may consider expedient including a condition

(a) that the land not be subdivided except for

(i) use for the purposes of agricultural, forestry or fisheries production,

(ii) conservation use pursuant to a designation as a natural area under the *Natural Areas Protection Act* R.S.P.E.I. 1988, Cap. N-2, an archaeological site under the Archaeological Sites Protection Act R.S.P.E.I. 1988, Cap. A-17 or a heritage place

under the Heritage Places Protection Act Stats. P.E.I. 1992, c.31,

(iii) parks use by the Federal, provincial or a municipal government,

(iv) residential use by the owner;

(b) that the land be identified under the land identification program;

(c) that the land be consolidated with an adjoining parcel or parcels of land;

(d) that the applicant become a resident within such period of time as may be specified in the permit.

LAND IDENTIFICATION PROGRAM

1. AUTHORITY

Where the sale or purchase of real property is subject to the approval of the Lieutenant Governor in Council pursuant to the Prince Edward Island Lands Protection Act R.S.P.E.I. 1988, L-5, the Lieutenant Governor in Council may make it a condition of consenting to the

Updated 2002

Prince Edward Island Lands Protection Act Forms Regulations

sale that the land be identified for non-development use pursuant to the Land Identification Regulations made under the Act.

2. Eligibility

Any parcel of land in excess of five (5) acres or having in excess of 165 feet of shore frontage that is the subject of an application pursuant to the Act may be required to be identified except where planning approval has been granted for development of the entire parcel.

Land identified for non-development use under a land identification agreement cannot be further subdivided or used for commercial or industrial purposes. IRAC 03/96

FORM 2

APPLICATION BY OR ON BEHALF OF A **CORPORATION** PURSUANT TO SECTION 5 OF THE PRINCE EDWARD ISLAND Lands Protection Act

CORPORATE APPLICANTS MAY BE REQUIRED TO ENTER INTO A LAND IDENTIFICATION AGREEMENT.

NOTE: Multiple parcels may be included on a single application form provided the transaction is between the same purchaser(s) and vendor(s). However, questions #10-19 should be answered for each parcel individually (attach a schedule if necessary).

1. Name of Corporation acquiring land holding:	
2. Place of incorporation:	
3. Permanent address:	
4. Type of business in which corporation is engage	ed:
5. If the parcel is being acquired through a trust ind	dicate:
Name of trustee:	
Address of trustee:	
6. Name of present land holder:	
7. Address of present land holder:	
8. Name of vendor (if different than present land o	
9. Address of vendor:	· · · · · · · · · · · · · · · · · · ·
10. Approximate date of acquisition by the PRESE	ENT LAND HOLDER:
11. Type of transaction (check one):	
(a) purchase (c) lease	
(b) bequest	
12 Property Number Acreage	Shore frontage
(from property tax bill)	(feet)
13. State the location of each parcel:	()
Community Township Lot C	ounty
14. State agreed purchase price:	
15. If the interest is being acquired by lease indicated	
16. Identify and describe the buildings and other st	
	-
17. Indicate the present acreage breakdown of each	n parcel separately:
(a) cropland (d) marsh	
(b) pasture (e) other	
(c) woodland	
18. State the intended use that is proposed for each	ı parcel:
(a) permanent residence (d) commercial/i	ndustrial
(b) seasonal residence (e) subdivision o	f lots
(c) agriculture (f) other	
19. Has each parcel been used for agriculture in the	e last five years?
20. If Executive Council imposes one or more co	onditions pursuant to section 9.(1) of the
Act (refer to Note on Page 2), are you prepared	to complete the transaction? If
ônoö, indicate reasons on a separate sheet.	-
21. If the applicant corporation is directly or indir	ectly controlled by any other corporation,
state the total aggregate land holdings in the provi	ince of the controlling corporation and its
subsidiaries (excluding the applicant corporation)	
22 If the applicant corporation directly or indire	ectly controls any other corporation state

22. If the applicant corporation directly or indirectly controls any other corporation, state the total aggregate land holdings in the province of its controlled corporations

7

23. For the applicant corporation list on a separate sheet and attach:

(a) parcel number and acreage of each parcel of land in the province now owned or leased by the corporation. The Act excludes any parcel that existed on May 1, 1995, of less than one acre that is located in a city or town;

(b) name and permanent address of officers and directors of the corporation;

(c) total number of voting shares issued by the applicant corporation;

(d) names, addresses and number of voting shares held by shareholders, both corporate and individual, owning more than 5% of the issued voting shares of the applicant corporation;

(e) for the shareholders listed in subsection (d) above list:

(i) parcel number and acreage of each parcel of land in the province now owned or leased by each shareholder (including holdings of minor children). The Act excludes any parcel that existed on May 1, 1995, of less than one acre that is located in a city or town; and

(ii) parcel number and acreage of land holding of any other corporation in which each shareholder owns more than 5% of the issued voting shares, total number of voting shares issued by that corporation and number of voting shares owned by that shareholder. The Act excludes any parcel that existed on May 1, 1995, of less than one acre that is located in a city or town

(f) if the applicant corporation owns more than 5% of the issued voting shares in any other corporation(s) but does not control the corporation(s), list the following for each corporation:

(i) percentage of issued voting shares owned by the applicant corporation;

(ii) parcel numbers and acreages of each parcel of land owned or leased;24. Indicate any other circumstances that are relevant

25. I hereby certify that the above information and attached legal description of the property is true and correct in all respects.

Signature of corporate officer or attorney

Date

.....

Name of Signatory (Please print)

Attach

(a) a legal description of the parcel to be acquired;

(b) an orthophoto showing the parcel to be acquired outlined in red;

(c) a cheque made payable to the Island Regulatory and Appeals Commission in the amount of \$500.00 or 1% of the agreed purchase price, whichever is greater (refer to Fees Regulations to determine if the fee is applicable).

RETURN COMPLETED FORM TO: Island Regulatory and Appeals Commission P. O. Box 577 Charlottetown, PE C1A 7N8 Telephone: 902-892-3501

NOTE: In accordance with section 9.(1) of the Lands Protection Act special conditions may be required. section 9.(1) states:

9.(1) The Lieutenant Governor in Council may impose such conditions on a permit issued under section 4 or 5 as he may consider expedient including a condition

(a) that the land not be subdivided except for

(i) use for the purposes of agricultural, forestry or fisheries production,

(ii) conservation use pursuant to a designation as a natural area under the *Natural Areas Protection Act* R.S.P.E.I. 1988, Cap. N-2, an archaeological site under the Archaeological Sites Protection Act R.S.P.E.I. 1988, Cap. A-17 or a heritage place under the Heritage Places Protection Act Stats. P.E.I. 1992, c.31,

(iii) parks use by the Federal, provincial or a municipal government,

(iv) residential use by the owner;

(b) that the land be identified under the land identification program;

(c) that the land be consolidated with an adjoining parcel or parcels of land;

(d) that the applicant become a resident within such period of time as may be specified in the permit.

LAND IDENTIFICATION PROGRAM

1. AUTHORITY

Cap. L-5

Where the sale or purchase of real property is subject to the approval of the Lieutenant Governor in Council pursuant to the Prince Edward Island Lands Protection Act R.S.P.E.I. 1988, L-5, the Lieutenant Governor in Council may make it a condition of consenting to the sale that the land be identified for non-development use pursuant to the Land Identification Regulations made under the Act.

2. Eligibility

Any parcel of land in excess of five (5) acres or having in excess of 165 feet of shore frontage that is the subject of an application pursuant to the Act may be required to be identified except where planning approval has been granted for development of the entire parcel.

Land identified for non-development use under a land identification agreement cannot be further subdivided or used for commercial or industrial purposes.

8

FORM 3

APPLICATION FOR A PERMIT TO HOLD AN AGGREGATE LAND HOLDING IN EXCESS OF THE LIMITS SET OUT IN SECTION 2 OF THE **PRINCE EDWARD ISLAND LANDS PROTECTION ACT**

Pursuant to Section 6.1 of the *P.E.I. Lands Protection Act* a person or corporation must apply for a permit to hold an aggregate land holding in excess of the limits.

1. Name of applicant person or corporation

2. Address for service

3. Amount of land (in acres) held in excess of the limit, for which a permit is sought

4. Attach a completed Form 4A or 4B (whichever is appropriate) or complete the Declaration Statement below.

5. I hereby certify that the information contained in this application is complete and true in all respects.

Signature of applicant or corporate officer

Date

This application may be based on the land holding information submitted to the Island Regulatory and Appeals Commission in 1995. In such cases the following declaration statement must be completed by the Applicant or in the case of a corporation, by an authorized signing officer of the corporation.

DECLARATION

I HEREBY DECLARE that the information provided in the form and appendices submitted to the Island Regulatory and Appeals Commission dated is an accurate and true disclosure of all of the Applicant's land holding as of December 31, 1994.

I DECLARE that the statements made and the information given therein are true in substance and in fact and I make this solemn declaration conscientiously believing it to be true, and knowing that it is of the same force and effect as if made under oath and by virtue of the *Canada Evidence Act*.

Declared by the said)
before me at	in the County	of)
	of, 20	
5	,	

A COMMISSIONER FOR TAKING AFFIDAVITS IN THE SUPREME COURT

Applicant/authorized signing officer of the corporation

FORM 4A

LAND HOLDING DISCLOSURE STATEMENT FOR A PERSON PURSUANT TO SECTION 6.1 OF THE P.E.I. LANDS PROTECTION ACT

INFORMATION SHOULD REFLECT LAND HOLDINGS AS OF DECEMBER 31, 1994

SECTION 1

Cap. L-5

1.1 Name of disclosing person	
1.2 Address for service	
If the disclosing person owns or leases land in the province lis	t the following information.
1.3 Report the total acreage owned (Appendix 1)	
1.4 Report the total acreage leased in (Appendix 2)	
1.5 Gross total land holding (Lines $1.3 + 1.4$)	
1.6Report the total acreage leased out (Appendix 3)	
1.7 Net total land holding (Lines 1.5 - 1.6)	
- ()	

SECTION 2

2. If the disclosing person owns more than 5% of the issued voting shares in any corporation, list the following information. (All information must be provided, unless prior consent to withhold information in columns 2, 3 and 4 has been granted by the Minister. Provide a copy of the Minister's consent.)

1	2	3	4
NAME OF CORPORATION			% of Issued Voting Shares OWNED BY DISCLOSING PERSON

2.2 List the following information for each corporation listed in Section 2.1.

1	2	3	4	5	6
NAME OF CORPORATION	TOTAL ACREAGE OWNED (FROM APPENDIX 4)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 5	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 6)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)

SECTION 3

If any corporation listed in Section 2.1 directly or indirectly controls or is directly or indirectly controlled by another corporation(s), list the following information for the other corporation(s). (Use a separate sheet for each corporation for Sections 3.1 to 3.5.) (Note that ôcorporation includes a partnership, cooperative association or body corporate whether formed or incorporated under the law of this province or any other province or of Canada or outside of Canada, and for the purposes of this Act a corporation and other corporations directly or indirectly controlled by the same person, group or organization shall be deemed to be one corporationö).

(Example Disclosing person owns more than 5% of the shares of Corporation 'A'. Corporation'A' controls Corporation 'B'. Corporation 'B' controls Corporation 'C'.

Updated 2002

Prince Edward Island Lands Protection Act Forms Regulations

Corporation 'C' controls Corporation 'D'. Corporations 'B', 'C'and 'D' will report in this Section.)

3.1 Name of the corporation

3.2 Address for service

- 3.5 Names and permanent addresses of all officers and directors.....

3.6 List the following information for the corporation(s) listed in Section 3.1.

1	2	3	4	5	6
NAME OF CORPORATION	TOTAL ACREAGE OWNED (FROM APPENDIX 7)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 8)	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 9)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)

SECTION 4

If the disclosing person is involved in a trust, either as a trustee or beneficiary, which owns land in the province, list the names of any other trustee(s) and beneficiaries and provide details as to whether the trustee(s) or beneficiaries have the right to present use, present occupation or present possession of the land. (Use a separate sheet for each trust for Sections 4.1 to 4.4.)

4.1 Name of the trust

4.2 Address for service 4.3 List the names and permanent addresses of all trustees and beneficiaries of the trust

4.4 Details as to whether the trustee(s) or beneficiaries have the right to present use, present occupation or present possession of the land

.....

4.5 List the following information for each trust listed in Section 4.1.						
1 2 3 4 5						
NAME OF TRUST	TOTAL ACREAGE OWNED (FROM APPENDIX 10)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 11)	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 12)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)	

SECTION 5

List the following information for any corporation (as defined in the Act) that has not been reported previously in this Form, that is directly or indirectly controlled, or directly or indirectly controls any other corporation listed in this Form. (Use a separate sheet for each corporation for Sections 5.1 to 5.5.)

5.1Name of the corporation

5.2Address for service

5.6 List the following information for each corporation listed in Section 5.1.

1	2	3	4	5	6
NAME OF CORPORATION	TOTAL ACREAGE OWNED (FROM APPENDIX 13)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 14	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 15)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)

The forms and appendices have been completed pursuant to Section 6.1 of the P.E.I. Lands Protection Act.

The onus will be on the Disclosing Person to provide sufficient evidence that the land holding listed in this form should be included in the Disclosing PersonÆs aggregate land holding as of December 31, 1994. Relevant supporting documentation including, but not necessarily limited to, deeds, leases, other agreements and written correspondence, along with any other evidence the Disclosing Person wishes considered, should be filed with this form.

If the Disclosing Person filed a Form 1 - Statement of Land Holdings for Persons under the P.E.I. Lands Protection Act, the Form 1 previously submitted shall be deemed to be an appendix to this form.

I HEREBY DECLARE that the information provided in this form and appendices is an accurate and true disclosure of all of my land holding as of December 31, 1994.

I DECLARE that the statements made and the information given herein are true in substance and in fact and I make this solemn declaration conscientiously believing it to be true, and knowing that it is of the same force and effect as if made under oath and by virtue of the Canada Evidence Act.

Declared by the said) before me at in the County of.......) and Province of .. this day of 20)

) Declaring person

A Commissioner for taking affidavits in

the Supreme Court

If this form is completed by someone other than the Disclosing Person list the following: Name of Person who completed the form

)

FORM 4A APPENDIX 1

For each parcel of land in the province owned by the disclosing person list the following information.

PARCELS OWNED BY THE DISCLOSING PERSON							
PARCEL NAME OF LOCATION ACREAGE TOTAL*							
NUMBER	REGISTERED						
	OWNER*						

Updated 2002 Prince Edward Island Lands Protection Act Forms Regulations

Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

FORM 4A APPENDIX 2

For each parcel of land in the province leased in by the disclosing person list the following information.

PARCELS LEASED IN BY THE DISCLOSING PERSON						
PARCEL	LEASE	ACREAGE				
NUMBER		PARCEL	FROM	TERM	LEASED IN	
			(OWNER)			

FORM 4A APPENDIX 3

If any of the parcels listed in Appendix 1 or 2 are leased out list the following information.

PARCELS LEASED OUT BY DISCLOSING PERSON				
PARCEL NUMBER	LEASED TO	LEASE TERM	ACREAGE LEASED OUT	

FORM 4A APPENDIX 4

List the following information for each parcel owned by the corporations listed in Section 2 (Complete a separate Appendix 4 for each corporation).

Name of corporation If any of the parcels listed in Appendix 1 or 2 are leased out list the following information.

PARCELS OWNED BY EACH CORPORATION				
PARCEL NUMBER	NAME OF REGISTERED OWNER*	LOCATION	ACREAGE	

*Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

FORM-4A APPENDIX-5

List the following information for each parcel leased in by corporations listed in Section 2 (Complete a separate Appendix 5 for each corporation).

Name of corporation

PARCELS LEASED IN BY EACH CORPORATION					
PARCEL NUMBER	LOCATION	TOTAL PARCEL ACREAGE	LEASED FROM (OWNER)	LEASE TERM	ACREAGE

Updated 2002

FORM 4A **APPENDIX 6**

List the following information for each parcel of land owned by corporation(s) listed in Section 3 (Complete a separate Appendix 7 for each corporation) Name of corporation

PARCELS LEASED OUT BY EACH CORPORATION				
PARCEL NUMBER LEASED TO LEASE TERM ACREAGE LEASED OUT				

FORM 4A **APPENDIX 7**

List the following information for each parcel of land owned by corporation(s) listed in Section 3 (Complete a separate Appendix 7 for each corporation) Name of corporation

PARCELS LEASED OUT BY EACH CORPORATION						
PARCEL NUMBER NAME OF REGISTERED OWNER LOCATION ACREAGE						

FORM 4A APPENDIX 8

List the following information for each parcel of land owned by corporation(s) listed in Section 3 (Complete a separate Appendix 7 for each corporation) Name of corporation

PARCELS LEASED OUT BY EACH CORPORATION				
PARCEL NUMBER	LOCATION	TOTAL PARCEL ACREAGE	LEASED FROM (OWNER)	ACREAGE LEASED IN

Initials

FORM 4A **APPENDIX 9**

If any of the parcels in Appendix 7 or 8 are leased out list the following information (Complete a separate Appendix 9 for each corporation). Name of corporation

PARCELS LEASED OUT BY EACH CORPORATION			
PARCEL NUMBER	LEASED TO	LEASE	ACREAGE
		TERM	LEASED OUT

Updated 2002

FORM 4A **APPENDIX 10**

List the following information for each parcel owned by trust(s) listed in Section 4 (Complete a separate Appendix 10 for each trust). Name of the trust

PARCELS OWNED BY EACH TRUST				
PARCEL NUMBER	NAME OF REGISTERED OWNER*	NAME OF BENEFICARY*	LOCATION	ACREAGE

FORM 4A **APPENDIX 11**

List the following information for each parcel owned by trust(s) listed in Section 4 (Complete a separate Appendix 10 for each trust). Name of the trust

PARCELS LEASED IN BY EACH TRUST					
PARCEL NUMBER	LOCATION	TOTAL PARCEL ACREAGE	LEASED FROM (OWNER)	LEASE TERM	ACREAGE LEASED IN

FORM 4A **APPENDIX 12**

If any of the parcels in Appendix 10 or 11 are leased out list the following information (Complete a separate Appendix 12 for each trust).

Name of the trust

PARCELS LEASED OUT BY EACH TRUST			
PARCEL NUMBER	LEASED TO	LEASE TERM	ACREAGE LEASED OUT

FORM 4A **APPENDIX 13**

List the following information for each parcel of land owned by corporation(s) listed in Section 5 (Complete a separate Appendix 13 for each corporation).

Name of corporation

PARCELS OWNED BY EACH CORPORATION			
PARCEL NUMBER	NAME OF REGISTERED	LOCATION	ACREAGE

Cap. L-5

15

OWNER*	

FORM 4A APPENDIX 14

List the following information for each parcel of land leased in by corporation(s) listed in Section 5 (Complete a separate Appendix 14 for each corporation).

Name of corporation

Cap. L-5

	PARCEI	LS LEASED IN B	Y EACH CORPO	RATION	
PARCEL NUMBER	LOCATION	TOTAL PARCEL	ACREAGEL EASED FROM (OWNER)	LEASE TERM	ACREAGE LEASED IN

FORM 4A APPENDIX 15

If any of the parcels in Appendix 13 or 14 are leased out list the following information (Complete a separate Appendix 15 for each corporation).

Name of corporation

PARCELS LEASED OUT BY EACH CORPORATION						
PARCEL NUMBER	LEASED TO	LEASE TERM	ACREAGE LEASED OUT			

FORM 4B LAND HOLDING DISCLOSURE STATEMENT FOR A CORPORATION PURSUANT TO SECTION 6.1 OF THE P.E.I. LANDS PROTECTION ACT

INFORMATION SHOULD REFLECT LAND HOLDINGS AS OF DECEMBER 31, 1994

SECTION 1

 1.1 Name of disclosing corporation

 1.2 Address for service

1.3 Place of incorporation

1.4 Date of incorporation

1.8 Gross total land holding

1.5 Names and permanent addresses of all officers and directors

If the disclosing corporation owns or leases land in the province list the following information.

1.6 Report the total acreage owned by the disclosing corporation (from Appendix 1)

.....

1.7 Report the total acreage leased in by the disclosing corporation (from Appendix 2)......

(Lines 1.6 + 1.7)

Updated 2002 Prince Edward Island Lands Protection Act Forms Regulations

Cap. L-5

1.9 Report the total acreage leased out by the disclosing corporation (from Appendix 3) 1.10 Net total land holding (Lines 1.8 - 1.9)

SECTION 2

2.1 Report the total number of voting shares issued by the disclosing corporation
2.2 List the names, addresses of the shareholders of the disclosing corporation and number of voting shares held by each shareholderùboth corporate and individualùowning more than 5% of the issued voting shares. (All information must be provided, unless prior consent to withhold information in columns 3 and 4 has been granted by the Minister. Provide a copy of the MinisterÆs consent.)

1	2	3	4
NAME OF SHAREHOLDER	ADDRESS	NUMBER OF SHARES OWNED BY SHAREHOLDER	% of Issued Voting Shares OWNED BY SHAREHOLDER

2.3 List the following information for each shareholder in Section 2.2.

1	2	3	4	5	6
NAME OF SHAREHOLDE R	TOTAL ACREAGE OWNED (FROM APPENDIX 4)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 5)	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 6)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)

If any shareholder listed in Section 2.2 holds more than 5% of the issued voting shares of any other corporation, list the following information. (Use a separate sheet for each shareholder for Sections 2.4 to 2.6.)

2.4 Name of shareholder

2.5 (All information must be provided, unless prior consent to withhold information in columns 3 and 4 has been granted by the Minister. Provide a copy of the Minister's consent.)

1	2	3	4
NAME OF CORPORATION	ADDRESS	NUMBER OF ISSUED SHARES OWNED BY SHAREHOLDER	% of Issued Voting Shares OWNED BY SHAREHOLDER

2.6 List the following information for each corporation in Section 2.5.

1	2	3	4	5	6
AME OF ORPORATIO	TOTAL ACREAGE OWNED (FROM APPENDIX 7)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 8)	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 9)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)

SECTION 3

Cap. L-5

If the disclosing corporation is controlled by another corporation and that controlling corporation indirectly controls any other corporation(s) or is directly or indirectly controlled by any other corporation(s), list the following for the other corporation(s). (Note that corporation includes a partnership, cooperative association or body corporate whether formed or incorporated under the law of this province or any other province or of Canada or outside of Canada, and for the purposes of this Act a corporation and other corporations directly or indirectly controlled by the same person, group or organization shall be deemed to be one corporation). Complete a separate Section 3 for each corporation.)

3.1 Name of the corporation

3.2 Address for service

3.3 Place of incorporation

3.4 Date of incorporation3.5 Names and permanent addresses of all officers and directors

3.6 List the following information for the corporation(s) listed in Section 3.1.

5.6 Elst the following information for the corporation(5) listed in Section 5.1.						
1	2	3	4	5	6	
NAME OF CORPORATIO N	TOTAL ACREAGE OWNED (FROM APPENDIX 10)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 11)	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 12)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)	

.....

3.7 List the following information for each shareholder who owns more than 5% of the issued voting shares in the corporation listed in 3.6 (Do not report a shareholder that reported in Section 2.2).

1	2	3	4	5	6
NAME OF	TOTAL	TOTAL	GROSS	TOTAL	NET TOTAL
SHAREHOLDE	ACREAGE	ACREAGE	TOTAL LAND	ACREAGE	LAND
R	OWNED	LEASED IN	HOLDING	LEASED OUT	HOLDING
	(FROM	(FROM	(COLUMN 2 +	(FROM	(COLUMN 4 -
	APPENDIX 13)	APPENDIX 14)	3)	APPENDIX 15)	5)

SECTION 4

If the disclosing corporation directly or indirectly controls any other corporation(s), list the following for each controlled corporation. (Complete a separate Section 4 for each corporation.)

- 4.1 Name of the corporation
- 4.2 Address for service
- 4.3 Place of incorporation

4.4 Date of incorporation

4.5 Names and permanent addresses of all officers and directors

4.6 List the following information for the corporation(s) listed in Section 4.1.

NAME OF TOTAL TOTAL GROSS TOTAL NE	TOTAL	

Updated 2002 Prince Edward Island Lands Protection Act Forms Regulations

CORPORATIO N	ACREAGE OWNED (FROM APPENDIX 16)	ACREAGE LEASED IN (FROM APPENDIX 17)	TOTAL LAND HOLDING (COLUMN 2 + 3)	ACREAGE LEASED OUT (FROM APPENDIX 18)	LAND HOLDING (COLUMN 4 - 5)

4.7 List the total number of voting shares issued by each corporation listed in 4.6 and the shareholders who own more than 5% of the issued voting shares of each corporation listed in 4.6. (All information must be provided, unless prior consent to withhold information in columns 2, 3 and 4 has been granted by the Minister. Provide a copy of the Minister's consent.)

1	2	3	4
NAME OF SHAREHOLDERS,	TOTAL NUMBER OF	UMBER OF ISSUED	% of Issued Voting
OTHER THAN THE	VOTING SHARES	VOTING SHARES	Shares OWNED
DISCLOSING CORPORATION,	ISSUED BY THE OTHER	OWNED BY EACH	BY EACH
WHO OWN MORE THAN 5%	CORPORATION	SHAREHOLDER	SHAREHOLDER
OF THE ISSUED VOTING			
SHARES OF THE			
CONTROLLED CORPORATION			

4.8 List the following information for each shareholder in Section 4.7

1	2	3	4	5	6
NAME OF EACH SHAREHOLDER	TOTAL ACREAGE OWNED (FROM APPENDIX 19)	TOTAL ACREAGE LEASED IN (FROM APPENDIX 20)	GROSS TOTAL LAND HOLDING (COLUMN 2 + 3)	TOTAL ACREAGE LEASED OUT (FROM APPENDIX 21)	NET TOTAL LAND HOLDING (COLUMN 4 - 5)

SECTION 5

5.1 If any of the shareholdings listed in this form are held in trust, list the names of the trustee(s) and beneficial owner(s) and provide details as to whether the trustee(s) or beneficiaries have present control of the shares. (Use a separate sheet for each trust.)5.2 Name of the trust

5.3 Address

5.4 List the names and permanent addresses of all trustees and beneficiaries of the trust

5.5 Provide details as to whether the trustee(s) or beneficiaries have present control of the shares

SECTION 6

The aggregate land holding of the disclosing corporation includes the land holdings of its shareholders (pursuant to Section 1.(1)(b)(ii) of the Act) and other corporations (as defined in Section 1.(1)(d) of the Act). If there are such shareholders or corporations that are not reflected in prior sections of this Form, provide the applicable information of shareholdings and land holding. (Use a separate sheet for each corporation.)

6.1 Name of the corporation

19

6.2 Adress for	service				
6.3 Pace of in	6.3 Pace of incorporation				
	6.4 Date of incorporation				
	6.5 Names and permanent addresses of all officers and directors				
	*				
••••••					
6.6 List the fo	llowing informa	tion for the corp	oration listed ir	Section 6.1.	
NAME OF	TOTAL	TOTAL	GROSS	TOTAL	NET TOTAL
CORPORATIO	ACREAGE	ACREAGE	TOTAL LAND	ACREAGE	LAND
N	OWNED	LEASED IN	HOLDING	LEASED OUT	HOLDING
	(FROM	(FROM	(COLUMN 2 +	(FROM	(COLUMN 4 -
	APPENDIX 22)	APPENDIX 23)	3)	APPENDIX 24)	5)

6.7 List the following information for each shareholder who owns more than 5% of the issued voting shares in the corporation listed in 6.6.

1	2	3	4	5	6
NAME OF	TOTAL	TOTAL	GROSS	TOTAL	NET TOTAL
EACH	ACREAGE	ACREAGE	TOTAL LAND	ACREAGE	LAND
SHAREHOLDE	OWNED	LEASED IN	HOLDING	LEASED OUT	HOLDING
R	(FROM	(FROM	(COLUMN 2 +	(FROM	(COLUMN 4 -
	APPENDIX 25)	APPENDIX 26)	3)	APPENDIX 27)	5)

The forms and appendices have been completed pursuant to Section 6.1 of the *P.E.I. Lands Protection Act.*

The onus will be on the Disclosing Corporation to provide sufficient evidence that the land holding listed in this form should be included in the Disclosing CorporationÆs aggregate land holding as of December 31, 1994. Relevant supporting documentation including, but not necessarily limited to, deeds, leases, other agreements and written correspondence, along with any other evidence the Disclosing Corporation wishes considered, should be filed with this form.

If the Disclosing Corporation filed a Form 2 - Statement of Land Holdings for Corporations under the P.E.I. Lands Protection Act, the previously filed Form 2 shall be deemed to be an appendix to this form.

I HEREBY DECLARE that the information provided in this form and appendices is an accurate and true disclosure of all of the Disclosing CorporationÆs land holding as of December 31, 1994.

I DECLARE that the statements made and the information given herein are true in substance and in fact and I make this solemn declaration conscientiously believing it to be true, and knowing that it is of the same force and effect as if made under oath and by virtue of the *Canada Evidence Act*.

Declared by the said)
(officer for the Disclosing C	orporation) before me)
at	in the County of)
)
and Province of)
thisday of	, 20 .)

Updated 2002

21

Applicant/authorized signing officer of the corporation

A COMMISSIONER FOR TAKING AFFIDAVITS IN THE SUPREME COURT

For inquiries please name contact person

Name

Address

Telephone

FORM 4B APPENDIX 1

List the following information for each parcel of land in the province owned by the disclosing corporation.

PARCELS OWNED BY DISCLOSING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 2 List the following information for each parcel of land in the province leased in by the disclosing corporation.

PARCELS LEASED IN BY DISCLOSING CORPORATIONPARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 3

If any of the parcels listed in Appendix 1 or 2 are leased out list the following information.

PARCELS LEASED OUT BY DISCLOSING CORPORATIONPARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 4

List the following information for each parcel owned by each shareholder listed in Section 2.2 (Complete a separate Appendix 4 for each shareholder).

Updated 2002

Name of Shareholder

Cap. L-5

PARCELS OWNED BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 5

List the following information for each parcel leased in by each shareholder listed in Section 2.2 (Complete a separate Appendix 5 for each shareholder).

Name of Shareholder

PARCELS LEASED IN BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 6

If any of the parcels listed in Appendix 4 or 5 are leased out list the following information (Complete a separate Appendix 6 for each shareholder).

Name of Shareholder

PARCELS LEASED OUT BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 7

List the following information for each parcel owned by any corporation listed in Section 2.5 (Complete a separate Appendix 7 for each corporation).

Name of Corporation

PARCELS OWNED BY OTHER CORPORATION(S)PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 8 Updated 2002 Prince Edward Island Lands Protection Act Forms Regulations

Cap. L-5

23

List the following information for each parcel leased in by any corporation listed in Section 2.5 (Complete a separate Appendix 8 for each corporation).

Name of Corporation

PARCELS LEASED IN BY OTHER CORPORATION(S)PARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 9

If any of the parcels listed in Appendix 7 or 8 are leased out list the following information. (Complete a separate Appendix 9 for each corporation).

Name of Corporation

PARCELS LEASED OUT BY OTHER CORPORATION(S)PARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 10

List the following information for each parcel of land owned by a corporation listed in Section 3.1.

PARCELS OWNED BY A CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 11

List the following information for each parcel of land leased in by a corporation listed in Section 3.1.

PARCELS LEASED IN BY A CORPORATIONPARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 12

If any of the parcels listed in Appendix 10 or 11 are leased out list the following information.

Updated 2002

PARCELS LEASED OUT BY A CORPORATION PARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 13

Cap. L-5

List the following information for each parcel owned by a shareholder (including holdings of minor children) listed in Section 3.7. (Complete a separate Appendix 13 for each shareholder).

Name of Shareholder

PARCELS OWNED BY SHAREHOLDER OF A CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 14

List the following information for each parcel leased in by a shareholder (including holdings of minor children) listed in Section 3.7. (Complete a separate Appendix 14 for each shareholder).

Name of Shareholder

PARCELS LEASED IN BY SHAREHOLDER OF A CORPORATIONPARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 15

If any of the parcels listed in Appendix 13 or 14 are leased out list the following information. (Complete a separate Appendix 15 for each shareholder).

Name of Shareholder

PARCELS LEASED OUT BY SHAREHOLDER OF A CORPORATIONPARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 16

List the following information for each parcel of land owned by each corporation listed in Section 4.1. (Complete a separate Appendix 16 for each corporation).

Updated 2002

25

Name of Corporation

PARCELS OWNED BY CONTROLLED CORPORATION(S)PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 17

List the following information for each parcel of land leased in by each corporation listed in Section 4.1. (Complete a separate Appendix 17 for each corporation).

Name of Corporation

PARCELS LEASED IN BY CONTROLLED CORPORATION(S)PARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 18

If any of the parcels listed in Appendix 16 or 17 are leased out list the following information (Complete a separate Appendix 18 for each corporation).

Name of Corporation

PARCELS LEASED OUT BY CONTROLLED CORPORATION(S)PARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 19

List the following information for each parcel of land owned by each shareholder listed in Section 4.7. (Complete a separate Appendix 19 for each shareholder).

Name of Shareholder

PARCELS OWNED BY EACH SHAREHOLDER OF CONTROLLED CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 20

List the following information for each parcel of land leased in by each shareholder listed in Section 4.7. (Complete a separate Appendix 20 for each shareholder).

Name of Shareholder

PARCELS LEASED IN BY EACH SHAREHOLDER OF CONTROLLED CORPORATIONPARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials

Cap. L-5

FORM 4B APPENDIX 21

If any of the parcels listed in Appendix 19 or 20 are leased out list the following information (Complete a separate Appendix 21 for each shareholder).

Name of Shareholder

PARCELS LEASED OUT BY EACH SHAREHOLDER OF CONTROLLED CORPORATIONPARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials

FORM 4B APPENDIX 22

List the following information for each parcel of land owned by each corporation listed in Section 6.1. (Complete a separate Appendix 22 for each corporation).

Name of Corporation

PARCELS OWNED BY CORPORATION(S)PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 23

List the following information for each parcel of land leased in by each corporation listed in Section 6.1. (Complete a separate Appendix 23 for each corporation).

Name of Corporation

PARCELS LEASED IN BY CORPORATION(S)PARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials FORM 4B APPENDIX 24 Updated 2002

27

If any of the parcels listed in Appendix 22 or 23 are leased out list the following information (Complete a separate Appendix 24 for each corporation).

Name of Corporation

PARCELS LEASED OUT BY CORPORATION(S)PARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials FORM 4B APPENDIX 25

List the following information for each parcel of land owned by each shareholder listed in Section 6.7. (Complete a separate Appendix 25 for each shareholder).

Name of Shareholder

PARCELS OWNED BY EACH SHAREHOLDER OF A CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 4B APPENDIX 26

List the following information for each parcel of land leased in by each shareholder listed in Section 6.7. (Complete a separate Appendix 26 for each shareholder).

Name of Shareholder

PARCELS LEASED IN BY EACH SHAREHOLDER OF A CORPORATIONPARCEL NUMBERLOCATIONTOTAL PARCEL ACREAGELEASED FROM (OWNER)LEASE TERMACREAGE LEASED INTOTAL

Initials

FORM 4B APPENDIX 27

If any of the parcels listed in Appendix 25 or 26 are leased out list the following information (Complete a separate Appendix 27 for each shareholder).

Name of Shareholder

PARCELS LEASED OUT BY EACH SHAREHOLDER OF A CORPORATIONPARCEL NUMBERLEASED TOLEASE TERMACREAGE LEASED OUTTOTAL

Initials

FORM 5A: SELF-CALCULATING LAND HOLDING AND DISCLOSURE STATEMENT FOR A PERSON

For Reporting Purposes Attach This Form To Form 4A

INFORMATION SHOULD REFLECT LAND HOLDINGS AS OF DECEMBER 31, 1994

Disclosing Person's Name:

Self-Calculated Gross Aggregate Land Holding:

Total acreage owned (Worksheet 5A, Table 4.1, Item E): acres

Total acreage leased in (Worksheet 5A, Table 4.1, Item J): _____ acres

Gross Aggregate Land Holding (Worksheet 5A, Table 4.1, Item K):______acres

Self-Calculated Net Aggregate Land Holding:

Total acreage leased out (Worksheet 5A, Table 4.2, Item E): acres

Net Aggregate Land Holding (Worksheet 5A, Table 4.2, Item F): ______acres

I HEREBY DECLARE that the information provided in this form is an accurate and true disclosure of all of my land holding as of December 31, 1994.

I DECLARE that the statements made and the information given herein are true in substance and in fact and I make this solemn declaration conscientiously believing it to be true, and knowing that it is of the same force and effect as if made under oath and by virtue of the Canada Evidence Act.

 Declared by the said
)

 (officer for the Disclosing Corporation) before me at
)

 in the County of
 and Province of

 this
 day of
 20 .

)
)
)

 Declared by the said
)

 in the County of
 and Province of

 Declared by the said
)

 Disclosing Person
)

A COMMISSIONER FOR TAKING AFFIDAVITS IN THE SUPREME COURT

Worksheet For Form 5A SELF-CALCULATING LAND HOLDING DISCLOSURE STATEMENT FOR A PERSON.

INFORMATION SHOULD REFLECT LAND HOLDINGS AS OF DECEMBER 31, 1994

The information listed in Form 4A will assist in completing the calculations in this worksheet.

Section 1: Basic Information

1.1 Name

1.2 Address

Section 2: RELEVANT LAND HOLDINGS ATTRIBUTED TO THE DISCLOSING PERSON BASED ON THE DISCLOSING PERSONÆS SHAREHOLDINGS IN A CORPORATION(S)

2.1 For <u>each</u> corporation in which the disclosing person owns more than 5% of the issued voting shares, complete the following table (use a separate sheet for each).

(AcreageA) Lands owned by each corporation (Column 2, Section 2.2 of Form 4A)B) Multiply A by disclosing person's % of issued voting shares in each corporation C) Lands leased in by each corporation (Column 3, Section 2.2 of Form 4A)D) Multiply C by disclosing person's % of issued voting shares in each corporationE) Add B + D = F) Lands leased out by each corporation (Column 5, Section 2.2 in Form 4A)G) Multiply F by the disclosing personÆs % of issued voting shares in each corporation H) E - G

Section 3: RELEVANT LAND Holdings ATTRIBUTED TO THE DISCLOSING PERSON BASED ON THE DISCLOSING PERSONÆS SHAREHOLDINGS IN CORPORATION(S) THAT ARE DIRECTLY OR INDIRECTLY CONTROLLED BY ANOTHER CORPORATION.

Note: For the purposes of the P.E.I. Lands Protection Act, a corporation and other corporations directly or indirectly controlled by the same person, group or organization shall be deemed to be one corporation.

(Example: Disclosing person owns more than 5% of the shares of Corporation ôAö. Corporation ôAö controls Corporation ôBö. Corporation ôBö controls Corporation ôCö. Corporation ôCö controls Corporation ôDö. Corporations ôBö, ôCö and ôDö will report in this Section.)

3.1 List the corporations that comprise the DEEMED CORPORATION other than corporations listed in Section 2.1

3.2 List the following for the corporations listed in Section 3.1.

PRIVATE AcreageA) Total land owned by corporation(s) (Column 2, Section 3.6 of Form 4A & Column 2, Section 5.6 of Form 4A)B) Determine the relevant amount (of A) to be attributed to the Disclosing Person C) Total land leased in by corporation(s) (Column 3, Section 3.6 of Form 4A & Column 3, Section 5.6 of Form 4A)D) Determine the relevant amount (of C) to be attributed to the Disclosing PersonE) Total land leased out by corporation(s) (Column 5, Section 3.6 of Form 4A & Column 5, Section 5.6 of Form 4A)F) Determine the relevant amount (of E) to be attributed to the Disclosing

4.1 Gross Aggregate Land Holding

Place the totals from the previous tables in this worksheet or Form 4A in the appropriate spaces below:

PRIVATE AcreageA) Total acreage owned by the disclosing person (Line 1.3 of Form 4A)F) Total acreage leased in by the disclosing person (Line 1.4 of Form 4A) B) Disclosing person's interest in land owned by a corporation (Table 2.1, Item B of this worksheet)G) Disclosing person's interest in land leased in by a corporation (Table 2.1, Item D of this worksheet)C) Disclosing person's interest in land owned by corporation(s) listed in Table 3.2, Item B of this worksheetH) Disclosing person's interest in land owned by a trust in which the disclosing person is a trustee or beneficiary of the trust (Column 2, Section 4.5 of Form 4A)I) Total land leased in by a trust in which the disclosing person is a trust of the trust (Column 3, Section 4.5 of Form 4A)E) Total acreage owned

 $\begin{array}{l} A+B+C+D=J) \mbox{ Total acreage leased in} \\ F+G+H+I=K) \mbox{ Gross Aggregate Land Holding} \\ E+J=4.2 \mbox{ Net Aggregate Land Holding} \end{array}$

4.2 Net Aggregate Land Holding

Place the totals from the previous tables in this Worksheet or in Form 4A in the appropriate spaces below:

PRIVATE AcreageA) Lands leased out by the disclosing person (Line 1.6 of Form 4A)B) Disclosing person's interest in land leased out by a corporation, each (Table 2.1, Item G of this worksheet)C) Disclosing person's interest in land leased out by controlling or controlled corporation, each (Table 3.2, Item F of this worksheet)D) Disclosing personÆs interest in land leased out by a trust, each (Column 5, Section 4.5 of Form 4A)E) Total acreage leased out

A + B + C + D = F) Gross aggregate land holding (Table 4.1, Item K) Minus total acreage leased out (Table 4.2, Item E)

Equals Net Aggregate Land Holding

FORM 5B

SELF-CALCULATING LAND AND HOLDING DISCLOSURE STATEMENT FOR A CORPORATION

For Reporting Purposes Attach This Form To Form 4B

INFORMATION SHOULD REFLECT LAND HOLDINGS AS OF DECEMBER 31, 1994

Updated 2002	Prince Edward Island Lands Protection Act	Cap. L-5
	Forms Regulations	

31

Disclosing Corporations Name:			
Self-Calculated Gross Aggregate Land Holding:			
Total acreage owned (Worksheet 5B, Table 6.1, Item J): acres			
Total acreage leased in (Worksheet 5B, Table 6.1, Item T):			
Gross Aggregate Land Holding (Worksheet 5B, Table 6.1, Item U):acres			
Self-Calculated Net Aggregate Land Holding:			
Total acreage leased out (Worksheet 5B, Table 6.2, Item J):			
Net Aggregate Land Holding (Worksheet 5B, Table 6.2, Item K):acres			
I HEREBY DECLARE that the information provided in this form is an accurate and true disclosure of all of my land holding as of December 31, 1994.			
I DECLARE that the statements made and the information given herein are true in substance and in fact and I make this solemn declaration conscientiously believing it to be true, and knowing that it is of the same force and effect as if made under oath and by virtue of the Canada Evidence Act.			

Declared by the s	aid)
(officer for the Di	isclosing Corp	oration) before me at)
in the County of		and Province of)
this	_day of	20 .)
))Authorized signing
)officer for the
)disclosing Corporation
)

A COMMISSIONER FOR TAKING AFFIDAVITS IN THE SUPREME COURT

Worksheet For Form 5B SELF-CALCULATING LAND HOLDING DISCLOSURE STATEMENT FOR A CORPORATION.

INFORMATION SHOULD REFLECT LAND HOLDINGS AS OF DECEMBER 31, 1994

The information listed in Form 4B will assist in completing the calculations in this worksheet.

Section 1: Basic Information

1.1 Name of disclosing corporation

1.2 Address

Section 2: Shareholders of disclosing corporation holdings through other corporations

2.1 For each shareholder listed in Section 2.4 of Form 4B, list the following (complete a separate table for each shareholder):

PRIVATE tAcreageA) Land owned by the other corporation(s) (Column 2, Section 2.6 of Form 4B) B) Multiply A by % of shares owned by each shareholder in the other corporation(s)C) Land leased in by the other corporation(s) (Column 3, Section 2.6 of Form 4B)D) Multiply C by % of shares owned by each shareholder in the other corporation(s)E) Land leased out by the other corporation(s) (Column 5, Section 2.6 of Form 4B)F) Multiply E by % of shares owned by each shareholder in the other corporation(s) (Column 5, Section 2.6 of Form 4B)F) Multiply E by % of shares owned by each shareholder in the other corporation(s) (Column 5, Section 2.6 of Form 4B)F) Multiply E by % of shares owned by each shareholder in the other corporation(s)

Section 3:

3.1 For each corporation listed in Section 3.1 of Form 4B, list the following (complete a separate table for each corporation):

PRIVATE AcreageA) Land owned by each corporation (Column 2, Section 3.6 of Form 4B)B) Land leased in by each corporation (Column 3, Section 3.6 of Form 4B)C) Land leased out by each corporation (Column 5, Section 3.6 of Form 4B)

3.2 For each shareholder listed in Section 3.7 of Form 4B, list the following (complete a separate table for each corporation):

PRIVATE AcreageA) Land owned by each shareholder (Column 2, Section 3.7 of Form 4B)B) Land leased in by each shareholder (Column 3, Section 3.7 of Form 4B)C) Land leased out by each shareholder (Column 5, Section 3.7 of Form 4B)

Section 4:

4.1 For each corporation listed in Section 4.1 of Form 4B, list the following (complete a separate table for each corporation):

PRIVATE AcreageA) Land owned by each corporation (Column 2, Section 4.6 of Form 4B)B) Land leased in by each corporation (Column 3, Section 4.6 of Form 4B)C) Land leased out by each corporation (Column 5, Section 4.6 of Form 4B)

4.2 For each shareholder listed in Section 4.8 of Form 4B, list the following (complete a separate table for each corporation):

PRIVATE AcreageA) Land owned by each shareholder (Column 2, Section 4.8 of Form 4B)B) Land leased in by each shareholder (Column 3, Section 4.8 of Form 4B)C) Land leased out by each shareholder (Column 5, Section 4.8 of Form 4B)Section 5:

Section 5:

5.1 For each corporation listed in Section 6.1 of Form 4B, list the following (complete a separate table for each corporation):

PRIVATE AcreageA) Land owned by each corporation (Column 2, Section 6.6 of Form 4B)B) Land leased in by each corporation (Column 3, Section 6.6 of Form 4B)C) Land leased out by each corporation (Column 5, Section 6.6 of Form 4B)

5.2 For each shareholder listed in Section 6.7 of Form 4B, list the following (complete a separate table for each corporation):

PRIVATE AcreageA) Land owned by each shareholder (Column 2, Section 6.7 of Form 4B)B) Land leased in by each shareholder (Column 3, Section 6.7 of Form 4B)C) Land leased out by each shareholder (Column 5, Section 6.7 of Form 4B)Section 6: Final Calculation

Section 6: FINAL CALCULATION

6.1 Gross Aggregate Land Holding

Place the totals from the previous tables in this Worksheet or Form 4B in the appropriate spaces below:

PRIVATE 0AcreageAcreageA) Total acreage owned by the disclosing corporation (1.6 of Form 4B)K) Total acreage leased in by the disclosing corporation (1.7 of Form 4B)B) Total acreage owned by the shareholders of disclosing corporation (Column 2, Section 2.3 of Form 4B)L) Total acreage leased in by the shareholders of disclosing corporation (Column 3, Section 2.3 of Form 4B)C) Total of Tables 2.1, Item B of this Worksheet)M) Total of Tables 2.1, Item D of this Worksheet)D) Totals of Tables 3.1, Item A of this WorksheetN) Totals of Tables 3.1, Item B of this WorksheetE) Totals of Tables 3.2, Item A of this WorksheetD) Totals of Tables 3.2, Item A of this WorksheetD) Totals of Tables 3.2, Item B of this WorksheetE) Totals of Tables 4.1, Item A of this WorksheetD) Totals of Tables 4.2, Item A of this WorksheetD) Totals of Tables 4.2, Item A of this WorksheetH) Totals of Tables 5.1, Item A of this WorksheetB) Totals of Tables 5.2, Item A of this WorksheetB) Totals of Tables 5.2, Item A of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) Totals of Tables 5.2, Item B of this WorksheetB) A+B+C+D+E+F+G+H+I =

T) K+L+M+N+O+P+Q+R+S=U) Gross Aggregate Land Holding J+T=

6.2 Net Aggregate Land Holdings:

Place the totals from the previous tables in this Worksheet or Form 4B in the appropriate spaces below:

PRIVATE rAcreageA) Land leased out by the disclosing corporation (Line 1.9 of Form 4B)B) Land leased out by shareholder (Column 5, Section 2.3 of Form 4B)C) Total of Tables 2.1, Item F of this Worksheet)D) Total of Tables 3.1, Item C of this Worksheet)E) Total of Tables 3.2, Item C of this Worksheet)F) Total of Tables 4.1, Item C of this Worksheet)G) Total of Tables 4.2, Item C of this Worksheet)H) Total of Tables 5.1, Item C of this Worksheet)I) Total of Tables 5.2, Item C of this Worksheet)J) A+B+C+D+E+F+G+H+I=K) Gross Aggregate Land Holding (Table 6.1, Item U of this Worksheet)

Minus total land leased out (Item J of this table)

Equals Net Aggregate Land Holding

FORM 6A LAND HOLDING DISCLOSURE STATEMENT FOR A PERSON PURSUANT TO SECTION 10 OF THE P.E.I. LANDS PROTECTION ACT

*****SHORT FORM*****

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

SECTION 1

Cap. L-5

Circle the answers to the following questions:

seq level0 *ALPHABETICB. Are there other corporations that own shares in a corporation in which you own more than 5% of the issued voting shares? YES NO

seq level0 *ALPHABETICC. Is the disclosing person a trustee or a beneficiary in a trust that has an aggregate land holding?

YES NO

seq level0 *ALPHABETICD. Has the Minister given the disclosing person approval to limit disclosure pursuant to subsection 11(3) of the **P.E.I. Lands Protection Act**? YES NO

If you answered YES to any of the above questions, do not use this form. Complete Form 6C.

Reporting Year

seq level0 \h \r0 seq level1 \h \r0 seq level2 \h \r0 seq level3 \h \r0 seq level4 \h \r0 seq level5 \h \r0 seq level5 \h \r0 seq level6 \h \r0 seq level7 \h \r0 1seq level0 *arabic1 Name of Disclosing Person

seq level0 \h \r0 seq level1 \h \r0 seq level2 \h \r0 seq level3 \h \r0 seq level4 \h \r0 seq level5 \h \r0 seq level6 \h \r0 seq level7 \h \r0 1seq level0 *arabic1 Mailing Address Province

Postal Code	
-------------	--

Telephone

Fax

If the Disclosing Person owns or leases land in the province list the following information.

1.3 Total acreage owned (Appendix 1)

1.4 Total acreage leased in (Appendix 2)

35

1.5 Corporate holdings from Section 2 (Total of Column 6 in Section 2.2)

1.6 Total Aggregate Land Holding (1.3 + 1.4 + 1.5)

Initials

SECTION 2

2.1 If the Disclosing Person owns more than 5% of the issued voting shares in any corporation, list the following information.

PRIVATE 1234NAME OF CORPORATIONTOTAL NUMBER OF ISSUED VOTING SHARESNUMBER OF ISSUED VOTING SHARES OWNED BY DISCLOSING PERSON% OF ISSUED VOTING SHARES

2.2 List the following information for each corporation listed in Section 2.1.

PRIVATE 123456NAME OF CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 3) TOTAL ACREAGE LEASED IN (FROM APPENDIX 4)TOTAL LAND HOLDING (COLUMN 2 +3)% OF ISSUED VOTING SHARES (FROM 2.1 COLUMN 4)CONTRIBUTION TO AGGREGATE LAND HOLDING (MULTIPLY COLUMN 4 BY COLUMN 5) TOTAL

Initials

This form has been completed pursuant to Section 10 of the Prince Edward Island Lands Protection Act.

I hereby declare that the information provided in this form and appendices is accurate and true to the best of my knowledge and belief for reporting year

Signature of Disclosing Person* Date

Name (Please Print)

Cap. L-5

* This form must be signed by the Disclosing Person.

For further inquiries please name a contact person

Name

Mailing Address

Province Postal Code

Telephone Fax

FORM 6A APPENDIX 1

For each parcel of land in the province owned by the Disclosing Person list the following information.

PRIVATE PARCELS OWNED BY THE DISCLOSING PERSONPRIVATE PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6A APPENDIX 2

For each parcel of land of land in the province leased in by the Disclosing Person list the following information.

PRIVATE PARCELS LEASED IN BY THE DISCLOSING PERSONPRIVATE PARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6A APPENDIX 3

List the following information for each parcel of land in the province owned by corporations listed in Section 2.1 (Complete a separate Appendix 3 for each corporation).

Name of corporation

PRIVATE PARCELS OWNED BY EACH CORPORATIONPRIVATE PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of

the registered owner.

Initials FORM 6A APPENDIX 4

List the following information for each parcel of land in the province leased in by corporations listed in Section 2.1. (Complete a separate Appendix 4 for each corporation.)

Name of corporation

PRIVATE PARCELS LEASED IN BY EACH CORPORATIONPRIVATE PARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials

FORM 6B LAND HOLDING DISCLOSURE STATEMENT FOR A CORPORATION PURSUANT TO SECTION 10 OF THE P.E.I. LANDS PROTECTION ACT

*****SHORT FORM*****

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

SECTION 1

Cap. L-5

Circle the answers to the following questions:

Does the disclosing corporation own more than 5% of the issued voting shares in any other corporations? YES NO Do any other corporations own more than 5% of the issued voting shares in the disclosing corporation? NO YES Do any of the shareholders of the disclosing corporation own more than 5% of the issued voting shares in any other corporations? YES NO Is the disclosing corporation or any of its shareholders a trustee or beneficiary in a trust that has an aggregate land holding? YES NO Has the Minister given the disclosing corporation approval to limit disclosure pursuant to subsection 11(3) of the P.E.I. Lands Protection Act? YES NO

If you answered YES to <u>any of the above questions</u>, do not use this form. Complete Form 6D.

Reporting Year

1.1	Name of Disclosing Corporation				
1.2	Mailing Address		Province		
	Postal Code	Telephone	Fax		
1.3	Place of incorporation	1.4	Date of incorporation		

1.5 Names and permanent addresses of all officers and directors

List the following information.

- 1.6 Total acreage owned by the Disclosing Corporation (from Appendix 1)
- 1.7 Total acreage leased in by the Disclosing Corporation (from Appendix 2)

Total land holdings of shareholders of the Disclosing Corporation (Total of 1.8 Column 4

in Section 2.3)

1.9 Total Aggregate Land Holding (1.6+1.7+1.8)

Initials

SECTION 2

2.1 Report the total number of voting shares issued by the Disclosing Corporation

List the names and addresses of the shareholders and number of voting shares 2.2 held by each shareholderùboth corporate and individualùowning more than 5% of the issued voting shares of the Disclosing Corporation.

1234NAME OF SHAREHOLDERADDRESSNUMBER OF SHARES OWNED BY A SHAREHOLDER% of Issued Voting Shares 2.3

List the following information for each shareholder in Section 2.2.

1234NAME OF SHAREHOLDERTOTAL ACREAGE OWNED (FROM APPENDIX 3) TOTAL ACREAGE LEASED IN (FROM APPENDIX 4)TOTAL LAND HOLDING (COLUMN 2 + 3)TOTAL

Initials **SECTION 3**

3.1 Pursuant to Section 11.(1)(f) of the Act, list any transfers of 10% or more of the issued voting shares which have been made since the last disclosure.

123NAME OF CORPORATIONNAME OF THE SHAREHOLDER WHO ACQUIRED 10% OR MORE OF THE SHARES % OF SHARES ACQUIRED BY THE SHAREHOLDER SINCE LAST DISCLOSURE

Initials

FORM 6B

Cap. L-5

This form has been completed pursuant to Section 10 of the P.E.I. Lands Protection Act.

I hereby declare that the information provided in this form and appendices is accurate and true to the best of my knowledge and belief for reporting year

Signature on behalf of the Disclosing Corporation* Date

This form must be signed by an authorized signing officer of the Disclosing Corporation.

Name of Signatory

Title

For further inquiries please name a contact person

Name

Mailing Address

Province Postal Code

Telephone Fax

FORM 6B APPENDIX 1

List the following information for each parcel of land in the province owned by the Disclosing Corporation.

PARCELS OWNED BY DISCLOSING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials

40

FORM 6B APPENDIX 2

List the following information for each parcel of land in the province leased in by the Disclosing Corporation.

PARCELS LEASED IN BY DISCLOSING CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6B APPENDIX 3

List the following information for each parcel of land in the province owned by each shareholder listed in Section 2.2. (Complete a separate Appendix 3 for each shareholder.)

Name of Shareholder

PARCELS OWNED BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6B APPENDIX 4

List the following information for each parcel of land in the province leased in by each shareholder listed in Section 2.2. (Complete a separate Appendix 4 for each shareholder.)

Name of Shareholder

PARCELS LEASED IN BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials

FORM 6C LAND HOLDING DISCLOSURE STATEMENT FOR A PERSON PURSUANT TO SECTION 10 OF THE P.E.I. LANDS PROTECTION ACT

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

SECTION 1

Cap. L-5

Reporting Year

1.1	Name of Disclosing Pe	rson		
1.2	Mailing Address	Province		
	Postal Code		Telephone	Fax

If the Disclosing Person owns or leases land in the province list the following information.

1.3	Total acreage owned (Appendix 1)
1.4	Total acreage leased in (Appendix 2)
1.5	Corporate holdings from Section 2 (Total of Column 6 in Section 2.2)
1.6	Corporate holdings from Section 3 (Total of Column 5 in Section 3.7)
1.7	Trust holdings from Section 4 (Total of Column 4 in Section 4.4)
1.8	Total Aggregate Land Holding (1.3+1.4+1.5+1.6+1.7)

Initials

2.1 If the Disclosing Person owns more than 5% of the issued voting shares in any corporation, list the following information. (All information must be provided, unless prior approval to withhold information in columns 2, 3 and 4 has been granted by the Minister. Provide a copy of Form 7A.)

1234NAME OF CORPORATIONTOTAL NUMBER OF ISSUED VOTING SHARESNUMBER OF ISSUED VOTING SHARES OWNED BY DISCLOSING PERSON% of Issued Voting Shares

2.2 List the following information for each corporation listed in Section 2.1.

123456NAME OF CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 3) TOTAL ACREAGE LEASED IN (FROM APPENDIX 4)TOTAL LAND HOLDING (COLUMN 2 +3)% of Issued Voting Shares (FROM 2.1 COLUMN 4)CONTRIBUTION TO AGGREGATE LAND HOLDING (MULTIPLY COLUMN 4 BY COLUMN 5) TOTAL

Initials SECTION 3

List the following information for any corporation (as defined in the Act) that is directly or indirectly controlled by any corporation listed in Section 2, but has not been reported in Section 2. (Use a separate sheet for each corporation for Sections 3.1 to 3.5.)

- 3.1 Name of the corporation
- 3.2 Address for service
- 3.3 Place of incorporation 3.4 Date of incorporation
- 3.5 Names and permanent addresses of all officers and directors

3.6 List the following information for the corporation(s) listed in Section 3.1. (All information must be provided, unless prior approval to withhold information in columns 2, 3 and 4 has been granted by the Minister. Provide a copy of Form 7A.)

12345NAME OF CORPORATION LISTED IN 3.1NAME OF SHAREHOLDER (CORPORATIONS LISTED IN SECTION 2 ONLY)TOTAL NUMBER OF ISSUED 43

VOTING SHARESNUMBER OF ISSUED VOTING SHARES OWNED BY THE CORPORATION IN SECTION 2 % of Issued Voting Shares

3.7 List the following information for each corporation listed in Section 3.1.

12345NAME OF CORPORATION

LISTED IN 3.1TOTAL ACREAGE OWNED (FROM APPENDIX 5) TOTAL ACREAGE LEASED IN (FROM APPENDIX 6)TOTAL LAND HOLDING (COLUMN 2 + 3)RELEVANT AMOUNT TO BE ATTRIBUTED TO DISCLOSING PERSON (SEE EXAMPLE)TOTAL Example: The Disclosing Person areas 20% of the charge of Correction A. Correction A sume

The Disclosing Person owns 25% of the shares of Corporation A. Corporation A owns 75% of the shares of Corporation B (Controlled Corporation). Corporation B has a land holding of 1,000 acres. Relevant amount equals $1,000 \times 75\% \times 25\% = 187.50$ acres.

Initials SECTION 4

List the following information if the Disclosing Person is involved in a trust, either as a trustee or beneficiary, which owns land in the province. (Use a separate sheet for each trust for Sections 4.1 to 4.3.)

4.1 Name of the trust
4.2 Address for service
4.3 List the names and permanent addresses of all trustees and beneficiaries of the trust

4.4 List the following information for each trust listed in Section 4.1.

234NAME OF TRUSTTOTAL ACREAGE OWNED (FROM APPENDIX 7) TOTAL ACREAGE LEASED IN (FROM APPENDIX 8)TOTAL LAND HOLDING (COLUMN 2 +3)TOTAL

Initials

45

This form has been completed pursuant to Section 10 of the Prince Edward Island Lands Protection Act.

I hereby declare that the information provided in this form and appendices is accurate and true to the best of my knowledge and belief for reporting year

Signature of Disclosing Person* Date

Name (Please Print)

* This form must be signed by the Disclosing Person.

For further inquiries please name a contact person

Name

Mailing Address

Province Postal Code

Telephone Fax FORM 6C APPENDIX 1

For each parcel of land in the province owned by the Disclosing Person list the following information.

PARCELS OWNED BY THE DISCLOSING PERSONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6C APPENDIX 2

For each parcel of land in the province leased in by the Disclosing Person list the following information.

Prince Edward Island Lands Protection Act Forms Regulations

Updated 2002

PARCELS LEASED IN BY THE DISCLOSING PERSONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6C APPENDIX 3

Cap. L-5

List the following information for each parcel of land in the province owned by corporations listed in Section 2 (Complete a separate Appendix 3 for each corporation).

Name of corporation

PARCELS OWNED BY EACH CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6C APPENDIX 4

List the following information for each parcel of land in the province leased in by corporations listed in Section 2. (Complete a separate Appendix 4 for each corporation.)

Name of corporation

PARCELS LEASED IN BY EACH CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6C APPENDIX 5

List the following information for each parcel of land in the province owned by corporations listed in Section 3 (Complete a separate Appendix 5 for each corporation).

Name of corporation

PARCELS OWNED BY EACH CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6C APPENDIX 6

47

List the following information for each parcel of land in the province leased in by corporations listed in Section 3. (Complete a separate Appendix 6 for each corporation.)

Name of corporation

PARCELS LEASED IN BY EACH CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6C APPENDIX 7

List the following information for each parcel of land in the province owned by trusts listed in Section 4. (Complete a separate Appendix 7 for each trust.)

Name of the trust

PARCELS OWNED BY THE TRUSTPARCEL NUMBERNAME OF REGISTERED OWNER*NAME OF BENEFICARYLOCATIONACREAGETOTALNote: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6C APPENDIX 8

List the following information for each parcel of land in the province leased in by trusts listed in Section 4. (Complete a separate Appendix 8 for each trust.)

Name of the trust

PARCELS LEASED IN BY THE TRUSTPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

FORM 6D LAND HOLDING DISCLOSURE STATEMENT FOR A CORPORATION PURSUANT TO SECTION 10 OF THE P.E.I. LANDS PROTECTION ACT

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

SECTION 1

Cap. L-5

Reporting Year

1.1	Name of Disclosing Corporation				
1.2	Mailing Address	Province			
	Postal Code		Telepho	one	Fax
1.3	Place of incorporation		1.4	Date of incorporation	t
1.5	Names and permanent addresses of all officers and directors				

List the following information.

1.6	Total acreage owned by the Disclosing Corporation (from Appendix 1)
1.7	Total acreage leased in by the Disclosing Corporation (from Appendix 2)
1.8 Column 4	Total land holdings of shareholders of the Disclosing Corporation (Total of in Section 2.3)
1.9	Relevant amount of land owned by other corporations (Total of column 6 in Section 2.6)
1.10	Total land holding of controlling corporation (Column 4 in Section 3.6)
1.11 4	Total land holdings of controlling corporationÆs shareholders (Total of column in Section 3.7)
1.12 4.A.7)	Total land holdings of controlled corporations (Total of column 4 in Section
1.13	Total land holdings of controlled corporationÆs shareholders (Total of column 4 in Section 4.A.8)

49

1.14 Total land holdings of corporations listed in Section 4.B (Total of column 6 in Section 4.B.7)

1.15 Total land holdings of corporations listed in Section 5 (Total of column 4 in Section 5.6)

1.16 Total land holdings of shareholders listed in Section 5 (Total of column 4 in Section 5.7)

1.17 Total Aggregate Land Holding (1.6+1.7+1.8+1.9+1.10+1.11+1.12+1.13+1.14+1.15+1.16)

Initials SECTION 2

2.1 Report the total number of voting shares issued by the Disclosing Corporation

2.2 List the names and addresses of the shareholders and number of voting shares held by each shareholder both corporate and individual owning more than 5% of the issued voting shares of the Disclosing Corporation. Corporate shareholders that hold more than 50% of the issued voting shares in the Disclosing Corporation should not be reported in this section. Also do not report a corporation that controls the Disclosing Corporation. (All information must be provided unless prior approval to withhold information in columns 3 and 4 has been granted by the Minister. Provide a copy of Form 7B.)

1234NAME OF SHAREHOLDERADDRESSNUMBER OF SHARES OWNED BY A SHAREHOLDER% of Issued Voting Shares2.3 List the following information for each shareholder in Section 2.2.

1234NAME OF SHAREHOLDERTOTAL ACREAGE OWNED (FROM APPENDIX 3) TOTAL ACREAGE LEASED IN (FROM APPENDIX 4)TOTAL LAND HOLDING (COLUMN 2 + 3)TOTAL

Initials

If any shareholder of the Disclosing Corporation holds more than 5% of the issued voting shares of any other corporation, list the following information. (Use a separate sheet for each shareholder for Sections 2.4 to 2.6.)

2.4 Name of shareholder

2.5 List the following information for the shareholders named in Section 2.4. (All information must be provided unless prior approval to withhold information in columns 3 and 4 has been granted by the Minister. Provide a copy of Form 7B.)

1234NAME OF OTHER CORPORATIONADDRESSNUMBER OF ISSUED VOTING
 SHARES OWNED BY A SHAREHOLDER% of Issued Voting Shares
 2.6 List the following information for each corporation in Section 2.5.

123456NAME OF OTHER CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 5) TOTAL ACREAGE LEASED IN (FROM APPENDIX 6)TOTAL LAND HOLDING (COLUMN 2 + 3)% of Issued Voting Shares (FROM 2.5 COLUMN 4)CONTRIBUTION TO TOTAL AGGREGATE LAND HOLDING (MULTIPLY COLUMN 4 BY COLUMN 5) TOTAL

Initials

Cap. L-5

SECTION 3

If the Disclosing Corporation is directly or indirectly controlled by any other corporation, list the following for the controlling corporation. (Note that corporation includes a partnership, cooperative association or body corporate whether formed or incorporated under the law of this province or any other province or of Canada or outside of Canada, and for the purposes of this Act a corporation and other corporations directly or indirectly controlled by the same person, group or organization shall be deemed to be one corporation.ö)

- 3.1 Name of the controlling corporation
- 3.2 Address for service
- 3.3 Place of incorporation 3.4 Date of incorporation
- 3.5 Names and permanent addresses of all officers and directors

3.6 List the following information for the corporation listed in Section 3.1.

1234NAME OF CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 7) TOTAL ACREAGE LEASED IN (FROM

APPENDIX 8)TOTAL LAND HOLDING (COLUMN 2 + 3)

3.7 List the following information for each shareholder who owns more than 5% of the issued voting shares in the corporation listed in 3.6. (Do not report a land holding that was previously reported in this form.)

1234NAME OF SHAREHOLDERTOTAL ACREAGE OWNED (FROM APPENDIX 9) TOTAL ACREAGE LEASED IN (FROM APPENDIX 10)TOTAL LAND HOLDING (COLUMN 2 + 3)TOTAL

51

Initials

SECTION 4.A

If the Disclosing Corporation directly or indirectly controls any other corporation(s) list the following for each corporation. (Use a separate sheet for each corporation for Sections 4.A.1 to 4.A.5.)

ion

4.A.2 Address for service

4.A.3 Place of incorporation 4.A.4 Date of incorporation

4.A.5 Names and permanent addresses of all officers and directors

4.A.6 List the total number of voting shares issued by each controlled corporation and the number of voting shares of the controlled corporation owned by the Disclosing Corporation. (All information must be provided unless prior approval to withhold information in columns 2, 3 and 4 has been granted by the Minister. Provide a copy of Form 7B.)

1234NAME OF EACH CONTROLLED CORPORATIONTOTAL NUMBER OF ISSUED VOTING SHARESNUMBER OF ISSUED VOTING SHARES OWNED BY THE DISCLOSING CORPORATION% of Issued Voting Shares

4.A.7 List the following information for each corporation in Section 4.A.1

1234NAME OF EACH CONTROLLED CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 11) TOTAL ACREAGE LEASED IN (FROM APPENDIX 12)TOTAL LAND HOLDING (COLUMN 2 + 3)TOTAL

Initials

4.A.8 List the following information for each shareholderùboth corporate and individualùowning more than 5% of the issued voting shares in each controlled corporation. Do not include the Disclosing Corporation or shareholders listed in Section 2.2. (Complete a separate table for each controlled corporation.)

Cap. L-5 Prince Edward Island Lands Protection Act Forms Regulations

Name of controlled corporation

1234NAME OF SHAREHOLDERTOTAL ACREAGE OWNED (FROM APPENDIX 13) TOTAL ACREAGE LEASED IN (FROM APPENDIX 14)TOTAL LAND HOLDING (COLUMN 2 + 3)TOTAL

SECTION 4.B

If the Disclosing Corporation owns more than 5% of the issued voting shares in any other corporation(s) list the following for each corporation (DO NOT REPORT CORPORATIONS THAT WERE REPORTED IN SECTION 4.A). (Use a separate sheet for each corporation for Sections 4.B.1 to 4.B.5.)

4.B.2 Address for service

4.B.3 Place of incorporation 4.B.4 Date of incorporation

4.B.5 Names and permanent addresses of all officers and directors

Initials

4.B.6 List the total number of voting shares issued by each corporation listed in 4.B.1. (All information must be provided unless prior approval to withhold information in columns 2, 3 and 4 has been granted by the Minister. Provide a copy of Form 7B.)

1234NAME OF CORPORATIONTOTAL NUMBER OF ISSUED VOTING SHARESNUMBER OF ISSUED VOTING SHARES OWNED BY DISCLOSING CORPORATION% of Issued Voting Shares 4.B.7 List the following information for each corporation in Section 4.B.1

123456NAME OF CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 15) TOTAL ACREAGE LEASED IN (FROM APPENDIX 16)TOTAL LAND HOLDING (COLUMN 2 + 3)% of Issued Voting Shares (FROM 4.B.6 COLUMN 4)CONTRIBUTION TO TOTAL AGGREGATE LAND HOLDING (MULTIPLY COLUMN 4 BY COLUMN 5) TOTAL

Initials

SECTION 5

The aggregate land holding of the Disclosing Corporation includes the land holdings of its shareholders (pursuant to Section 1.(1)(b)(ii) of the Act) and other corporations that may be part of the deemed corporation as defined in Section 1.(1)(d) of the Act. If there are such shareholders or corporations that are not reported in prior sections of this form, provide the applicable information of shareholdings and land holdings. (Complete a separate sheet for each corporation.)

- 5.1 Name of the corporation
- 5.2 Address for service
- 5.3 Place of incorporation 5.4 Date of incorporation
- 5.5 Names and permanent addresses of all officers and directors

5.6 List the following information for the corporation listed in Section 5.1.

1234

NAME OF CORPORATIONTOTAL ACREAGE OWNED (FROM APPENDIX 17) TOTAL ACREAGE LEASED IN (FROM APPENDIX 18)TOTAL LAND HOLDING (COLUMN 2 + 3) 5.7 List the following information for each shareholder who owns more than 5% of the issued voting shares in the corporation listed in 5.6.

1234

NAME OF SHAREHOLDERTOTAL ACREAGE OWNED (FROM APPENDIX 19) TOTAL ACREAGE LEASED IN (FROM

Cap. L-5 Prince Edward Island Lands Protection Act Updated 2002 Forms Regulations

APPENDIX 20)TOTAL LAND HOLDING (COLUMN 2 + 3)TOTAL

Initials

SECTION 6

6.1 If any of the shareholdings listed in this form are held in trust, list the names of the trustee(s) and beneficial owner(s). (Use a separate sheet for each trust.)

- 6.2 Name of the trust
- 6.3 Address
- 6.4 Name of the Corporation in which the trust holds the shares
- 6.5 List the names and permanent addresses of all trustees and beneficiaries of the trust

55

Initials

SECTION 7

7.1 Pursuant to Section 11.(1)(f) of the Act, list any transfers of 10% or more of the issued voting shares which have been made since the last disclosure.

123NAME OF CORPORATIONNAME OF THE SHAREHOLDER WHO ACQUIRED 10% OR MORE OF THE SHARES % OF SHARES ACQUIRED BY THE SHAREHOLDER SINCE LAST DISCLOSURE

Initials

This form has been completed pursuant to Section 10 of the P.E.I. Lands Protection Act.

I hereby declare that the information provided in this form and appendices is accurate and true to the best of my knowledge and belief for reporting year

Signature on behalf of the Disclosing Corporation* Date

* This form must be signed by an authorized signing officer of the Disclosing Corporation.

Name of Signatory

Title

For further inquiries please name a contact person

Name

Mailing Address

Province Postal Code

Telephone Fax

FORM 6D APPENDIX 1

List the following information for each parcel of land in the province owned by the Disclosing Corporation which includes lands held in trust where the Disclosing Corporation is a trustee or beneficiary of the trust.

PARCELS OWNED BY DISCLOSING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 2

List the following information for each parcel of land in the province leased in by the Disclosing Corporation.

PARCELS LEASED IN BY DISCLOSING CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

57

Initials FORM 6D APPENDIX 3

List the following information for each parcel of land in the province owned by each shareholder listed in Section 2.2. (Complete a separate Appendix 3 for each shareholder.)

Name of Shareholder

PARCELS OWNED BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 4

List the following information for each parcel of land in the province leased in by each shareholder listed in Section 2.2. (Complete a separate Appendix 4 for each shareholder.)

Name of Shareholder

PARCELS LEASED IN BY SHAREHOLDER(S) OF DISCLOSING CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D APPENDIX 5

List the following information for each parcel of land in the province owned by each corporation listed in Section 2.5. (Complete a separate Appendix 5 for each corporation.)

Name of Corporation

PARCELS OWNED BY OTHER CORPORATION(S)PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 6

List the following information for each parcel of land in the province leased in by each corporation listed in Section 2.5. (Complete a separate Appendix 6 for each corporation.)

Name of Corporation

PARCELS LEASED IN BY OTHER CORPORATION(S)PARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D APPENDIX 7

Cap. L-5

List the following information for each parcel of land in the province owned by the corporation listed in Section 3.1.

PARCELS OWNED BY THE CONTROLLING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL*

Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 8

List the following information for each parcel of land in the province leased in by the corporation listed in Section 3.1.

PARCELS LEASED IN BY THE CONTROLLING CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D APPENDIX 9

List the following information for each parcel of land in the province owned by a shareholder listed in Section 3.7. (Complete a separate Appendix 9 for each shareholder.)

Name of Shareholder

PARCELS OWNED BY SHAREHOLDER OF A CONTROLLING CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 10

Prince Edward Island Lands Protection Act Forms Regulations

List the following information for each parcel of land in the province leased in by a shareholder listed in Section 3.7. (Complete a separate Appendix 10 for each shareholder.)

Name of Shareholder

PARCELS LEASED IN BY SHAREHOLDER OF A CONTROLLING CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D APPENDIX 11

List the following information for each parcel of land in the province owned by each corporation listed in Section 4.A.1. (Complete a separate Appendix 11 for each corporation.)

Name of Corporation

PARCELS OWNED BY CONTROLLED CORPORATION(S)PARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 12

List the following information for each parcel of land in the province leased in by each corporation listed in Section 4.A.1. (Complete a separate Appendix 12 for each corporation.)

Name of Corporation

PARCELS LEASED IN BY CONTROLLED CORPORATION(S)PARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D APPENDIX 13

List the following information for each parcel of land in the province owned by each shareholder listed in Section 4.A.8. (Complete a separate Appendix 13 for each shareholder.)

59

Prince Edward Island Lands Protection Act Forms Regulations

Updated 2002

Name of Shareholder

Cap. L-5

PARCELS OWNED BY EACH SHAREHOLDER OF CONTROLLED CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 14

List the following information for each parcel of land in the province leased in by each shareholder listed in Section 4.A.8. (Complete a separate Appendix 14 for each shareholder.)

Name of Shareholder

PARCELS LEASED IN BY EACH SHAREHOLDER OF CONTROLLED CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D APPENDIX 15

List the following information for each parcel of land in the province owned by each corporation listed in Section 4.B.1. (Complete a separate Appendix 15 for each corporation.)

Name of Corporation

PARCELS OWNED BY EACH CORPORATION IN WHICH DISCLOSING CORPORATION OWNS MORE THAN 5% OF SHARESPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 16

List the following information for each parcel of land in the province leased in by each corporation listed in Section 4.B.1. (Complete a separate Appendix 16 for each corporation.)

Name of Corporation

PARCELS LEASED IN BY EACH CORPORATION IN WHICH DISCLOSING CORPORATION OWNS MORE THAN 5% OF SHARESPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE

M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D **APPENDIX 17**

List the following information for each parcel of land in the province owned by each corporation listed in Section 5.6 (Complete a separate Appendix 17 for each corporation).

PARCELS OWNED BY A CORPORATIONPARCEL NUMBERNAME OF REGISTERED OWNER*LOCATIONACREAGETOTAL* Note: If land is owned jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D APPENDIX 18

List the following information for each parcel of land in the province leased in by each corporation listed in Section 5.6 (Complete a separate Appendix 18 for each corporation).

PARCELS LEASED IN BY A CORPORATIONPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials FORM 6D **APPENDIX 19**

List the following information for each parcel of land in the province owned by each shareholder listed in Section 5.7 (Complete a separate Appendix 19 for each corporation).

PARCELS OWNED BY A SHAREHOLDERPARCEL NUMBERNAME OF Note: If land is owned REGISTERED OWNER*LOCATIONACREAGETOTAL* jointly or as tenants in common list the complete name of the registered owner.

Initials FORM 6D **APPENDIX 20**

List the following information for each parcel of land in the province leased in by each shareholder listed in Section 5.7 (Complete a separate Appendix 20 for each shareholder).

PARCELS LEASED IN BY A SHAREHOLDERPARCEL NUMBERLOCATIONLEASED FROMLEASE START DATE 61

M/D/YLEASE EXPIRY DATE M/D/YACREAGE LEASED INTOTAL

Initials

Cap. L-5

FORM 7A SELF-CALCULATING LANDHOLDING DISCLOSURE STATEMENTS FOR A PERSON For Reporting Purposes Attach This Form To Form 6C

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

Reporting Year

Disclosing Person's Name:

Self-Calculated Aggregate Land Holding:

Total acreage owned (Section 4.1, item E): ______acres

Total acreage leased in (Section 4.1, item J): ______ acres

Aggregate Land Holding (Section 4.1, item K): ______acres

Affirmation: I hereby declare that the information contained in this form is accurate and true to the best of my knowledge and belief.

Name:	

Signed*:

Date:

* This form must be signed by the Disclosing Person

MinisterÆs Approval

I am satisfied that it is not in the public interest for the shareholdings of individual shareholders to be disclosed pursuant to Subsection 11.(3) of the P.E.I. Lands Protection Act.

Minister's signature:WORKSHEET FOR FORM 7A SELF-CALCULATION OF AGGREGATE LNAD HOLDING FOR A PERSON.

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

The information listed in Form 6C will assist in completing the calculations in this worksheet.

SECTION 1

Reporting Year

- 1.1 Name
- 1.2 Address

SECTION 2

2.1 For each corporation in which the Disclosing Person owns more than 5% of the issued voting shares, list the following information (use a separate sheet for each). HOWEVER, DO NOT INCLUDE corporations that are directly or indirectly controlled by any other corporation. LIST THAT INFORMATION IN SECTION 3.1.

PRIVATE 1 AcreageA) Lands owned by the corporation (Column 2, Section 2.2 of Form 6C)B) Multiply A by Disclosing Person's percentage of issued voting shares in the corporation (relevant amount)C) Lands leased in by the corporation (Column 3, Section 2.2 of Form 6C)D) Multiply C by Disclosing Person's percentage of issued voting shares in the corporation (relevant amount)

SECTION 3

Note: For the purposes of the P.E.I. Lands Protection Act, a corporation and other corporations directly or indirectly controlled by the same person, group or organization shall be deemed to be one corporation.

3.1 For each corporation that is directly or indirectly controlled by another corporation and in which the Disclosing Person owns more than 5% of the issued voting shares list the following information.

PRIVATE AcreageA) Land owned by the controlled corporation(s) (Column 2, Section 3.7 of Form 6C)B) Lands leased in by the controlled corporation(s) (Column 3, Section 3.7 of Form 6C)C) Relevant amount* of land owned by controlled corporation(s) based on the

shareholdingsD) Relevant amount* of land leased in by controlled corporation(s) based on the shareholdings

* Pursuant to Subsection 1.(2) of the Act and shown calculated in Section 3.7 of Form 6C

SECTION 4

Cap. L-5

4.1 Aggregate Land Holding

Please place the totals from the previous sections in this worksheet or from Form 6C in the appropriate lines below:

PRIVATE AcreageAcreageA) Total acreage owned by the Disclosing Person (Line 1.3 of Form 6C)F) Total acreage leased in by the Disclosing Person (Line 1.4 of Form 6C) B) Disclosing PersonÆs relevant amount of land owned by corporation(s) listed in item B, Section 2.1 of this worksheet)G) Disclosing PersonÆs relevant amount of land leased in by corporation(s) listed in item D, Section 2.1 of this worksheetC) Disclosing PersonÆs relevant amount of land owned by controlled corporation(s) listed in item B, Section 3.1 of this worksheet)H) Disclosing PersonÆs relevant amount of land leased in by controlled corporation(s) listed in item D, Section 3.1 of this worksheetD) Total acreage owned by a trust in which the Disclosing Person is a trustee or beneficiary of the trust, each (Column 2, Section 4.4 of Form 6C)I) Total land leased in by a trust in which the Disclosing Person is a trustee or beneficiary of the trust, each (Column 3, Section 4.4 of Form 6C)E) Total acreage owned

A + B + C + D = J) Total acreage leased in

F + G + H + I = K Aggregate Land Holding =

FORM 7B SELF-CALCULATING LAND HOLDING DISCLOSURE STATEMENT FOR A

CORPORATION. For Reporting Purposes Attach This Form To Form 6D

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

Reporting Year

Disclosing CorporationÆs Name

Self-Calculated Aggregate Land Holding: Total acreage owned (Section 4.1, item J): acres Total acreage leased in (Section 4.1, item T): acres Aggregate Land Holding (Section 4.1, item U): acres

As an officer of the corporation named above, I hereby declare that the information contained in this form is accurate and true to the best of my knowledge and belief.

Name:

Title:

Signed*:

Date:

This form must be signed by an authorized signing officer of the Disclosing Corporation.

Minister's Approval

I am satisfied that it is not in the public interest for the shareholdings of individual shareholders to be disclosed pursuant to Subsection 11.(3) of the P.E.I. Lands Protection Act.

Minister's signature

65

WORKSHEET FOR FORM 7B SELF-CALCULATION OF AGGREGATE LAND HOLDING FOR A CORPORATION.

INFORMATION PROVIDED SHALL DISCLOSE THE MAXIMUM LAND HOLDING DURING THE REPORTING YEAR

The information listed in Form 6D will assist in completing the calculations in this worksheet.

SECTION 1

Cap. L-5

Reporting Year

1.1 Name of Disclosing Corporation

SECTION 2

2.1 For each shareholder who owns more than 5% of the issued voting shares of the Disclosing Corporation, who also owns more than 5% of the issued voting shares in any other corporation except one that is controlled by or controls the Disclosing Corporation, list the following (attach a separate sheet for each shareholder):

PRIVATE tAcreageA) Land owned by the other corporation (Column 2, Section 2.6 of Form 6D) B) Multiply A by percentage of shares owned by the other corporation() Land leased in by the other corporation (Column 3, Section 2.6 of Form 6D)D) Multiply C by percentage of shares owned in the other corporation

SECTION 3

3.1 For each corporation in which the Disclosing Corporation owns more than 5% of the issued voting shares but is not directly or indirectly controlled by the Disclosing Corporation, please list the following: (Complete A through D separately for each corporation, attach separate sheets if required.)

PRIVATE AcreageA) Land owned by each corporation (Column 2, Section 4.B.7 of Form 6D)B) Multiply A by Disclosing Corporation's percentage of shares (Column 4, Section 4.B.6 of Form 6D)C) Land leased in by each corporation (Column 3, Section 4.B.7 of Form 6D)D) Multiply C by Disclosing Corporation's percentage of shares (Column 4, Section 4.B.6 of Form 6D)WORKSHEET FOR FORM 7B

SECTION 4

4.1 Aggregate Land Holding

Place the totals from the previous sections in this Worksheet or Form 6D in the appropriate spaces below:

PRIVATE 0AcreageAcreageA) Total acreage owned by the Disclosing Corporation (Line 1.6 of Form 6D)K) Total acreage leased in by the Disclosing Corporation (Line 1.7 of Form 6D)B) Total acreage owned by the Disclosing CorporationÆs shareholders (Column 2, Section 2.3 of Form 6D)L) Total acreage leased in by the Disclosing CorporationÆs shareholders (Column 3, Section 2.3 of Form 6D)C) Relevant amount of land owned by other corporations attributed to the Disclosing Corporation (Section 2.1, Item B of worksheet 7B)M) Relevant amount of land leased in by other corporations attributed to the Disclosing Corporation Section 2.1, Item D of worksheet 7B)D) Total acreage owned by

controlling corporation (Column 2, Section 3.6 of Form 6D)N) Total acreage leased in by controlling corporation (Column 3, Section 3.6 of Form 6D)E) Total acreage owned by controlled corporation(s) (Column 2, Section 4.A.7 of Form 6D)O) Total acreage leased in by controlled corporation(s) (Column 3, Section 4.A.7 of Form 6D)F) Total acreage owned by shareholders of controlling corporation (Column 2, Section 3.7 of Form 6D)P) Total acreage leased in by shareholders of controlling corporation (Column 3, Section 3.7 of Form 6D)G) Total acreage owned by shareholders of controlled corporation(s) - Do not include the Disclosing Corporation or shareholders listed in Section 2.2 of Form 6D (Column 2, Section 4.A.8 of Form 6D)Q) Total acreage leased in by shareholders of controlled corporation(s) - Do not include the Disclosing Corporation or shareholders listed in Section 2.2 of Form 6D (Column 3, Section 4.A.8 of Form 6D)H) Relevant amount of land owned by other corporation(s) in which the Disclosing Corporation is a minor shareholder (Item B, Section 3.1 of this worksheet)R) Relevant amount of land leased in by other corporation(s) in which the Disclosing Corporation is a minor shareholder (Item D, Section 3.1 of this worksheet)I) Total acreage owned by corporation(s) listed in Section 5.7 of Form 6D (Column 2, Section 5 of

Form 6D)S) Total acreage leased in by corporation(s) listed in Section 5.7 of Form 6D (Column 2, Section 5 of Form 6D)J) A + B + C + D + E + F + G + H + I =T) K + L + M + N + O + P + Q + R + S =U) Aggregate Land Holding

FORM 8

APPLICATION BY OR ON BEHALF OF A **NON-RESIDENT OR CORPORATION** PURSUANT TO SUBSECTION 5.3(1)(b) OF THE *PRINCE EDWARD ISLAND LANDS PROTECTION ACT*

1. Name of applicant:

2. Citizenship (for applicant person):

3. Place of incorporation (for applicant corporation): Type of business in which applicant corporation is engaged:

4. Permanent address:

5. If land is being acquired through a trust indicate: Name of trustee: Address of trustee:

6. Total acreage proposed to be leased:

7. Intended use of proposed leased land:

PLEASE PROVIDE THE FOLLOWING INFORMATION IF THE APPLICANT IS A PERSON:

8. Indicate parcel number and acreage of each parcel of land in the province in which the applicant and his minor children hold an interest, including land held by way of lease. The Act excludes any parcel that existed on May 1, 1995 of less than one acre that is located in a city or town. (Attach separate sheet if necessary.)

Cap. L-5

9. For any corporation owning or leasing land within the province in which the applicant or his minor children hold more than 5% of the issued voting shares, list on a separate sheet and attach:

(a) name and permanent address of the corporation;

(b) total number of voting shares issued by the corporation;

(c) total number of issued voting shares owned by the applicant and his minor children; and

(d) parcel number and acreage of each parcel of land in the province now owned or leased by the corporation. The Act excludes any parcel that existed on May 1, 1995 of less than one acre that is located in a city or town.

PLEASE PROVIDE THE FOLLOWING INFORMATION IF THE APPLICANT IS A CORPORATION:

10. If the applicant corporation is directly or indirectly controlled by any other corporation, state the total aggregate land holdings in the province of the controlling corporation and its subsidiaries (excluding the applicant corporation)

11. If the applicant corporation directly or indirectly controls any other corporation, state the total aggregate land holdings in the province of its controlled corporations.

12. For the applicant corporation list on a separate sheet and attach:

(a) parcel number and acreage of each parcel of land in the province now owned or leased by the corporation. The Act excludes any parcel that existed on May 1, 1995 of less than one acre that is located in a city or town;

(b) name and permanent address of officers and directors of the corporation;

(c) total number of voting shares issued by the applicant corporation;

(d) names, addresses and number of voting shares held by shareholders, both corporate and individual, owning more than 5% of the issued voting shares of the applicant corporation;

(e) for the shareholders listed in subsection (d) above list:

(i) parcel number and acreage of each parcel of land in the province now owned or leased by each shareholder (including holdings of minor children). The Act excludes any parcel that existed on May 1, 1995 of less than one acre that is located in a city or town; and

(ii) parcel number and acreage of land holding of any other corporation in which each shareholder owns more than 5% of the issued voting shares, total number of voting shares issued by that corporation and number of voting shares owned by that shareholder. The Act excludes any parcel that existed on May 1, 1995 of less than one acre that is located in a city or town;

(f) if the applicant corporation owns more than 5% of the issued voting shares in any other corporation(s) but does not control the corporation(s), list the following for each corporation:

(i) percentage of issued voting shares owned by the applicant corporation; (ii) parcel numbers and acreage of each parcel of land owned or leased;

FOR APPLICANT PERSONS AND CORPORATIONS:

13. Indicate any other circumstances that are relevant

14. I hereby certify that the above information is true and correct in all respects.

Signature of applicant or his/her attorney

Date

Name of signatory (please print)

A fee of \$500.00 is applicable for an application filed by a non-resident person, a nonresident corporation or a corporation where the majority of shares are owned by nonresident(s).

Please attach a cheque made payable to the Island Regulatory and Appeals Commission in the amount of \$500.00 if the fee is applicable.

RETURN COMPLETED FORM TO:

Island Regulatory and Appeals Commission P. O. Box 577 Charlottetown PE C1A 7L1 Telephone: 902-892-3501

NOTE: In accordance with subsection 5.3(2) of the *Prince Edward Island Lands Protection Act* where permission has been granted by the Lieutenant Governor in Council for a lease of land, the person or corporation shall within one year of receiving permission and in every subsequent year, prior to December 31 file a statement disclosing information prescribed by subsection (3).

Subsections 5.3 (2) and (3) of the Act state:

(2) Where permission has been granted by the Lieutenant Governor in Council for a lease of land pursuant to an application made under clause (1)(b), the person or corporation shall

(a) within one year of receiving permission; and

(b) in every subsequent year, prior to December 31,

file a statement disclosing information prescribed by subsection (3).

(3) The statement required by subsection (2) shall disclose

(a) the parcel number;

(b) the acreage leased; and

(c) the term of the lease or leases,

for each parcel leased during the reporting period covered by the statement.

PLEASE SEE FORM 8, APPENDIX A STATEMENT FOR REPORTING LAND LEASED IN

FORM 8

APPENDIX A STATEMENT FOR REPORTING LAND LEASED IN PURSUANT TO SUBSECTION 5.3(2) OF THE PRINCE EDWARD ISLAND LANDS PROTECTION ACT

Please report all land leased in <u>during</u> the past calendar year. List the following information for each parcel.

Name of Lessee:

Prince Edward Island Lands Protection Act Forms Regulations

Updated 2002

LEASED LAND						
(1)	(2)	(3)	(4)		(5)	(6)
Parcel #	Location	Acres	Leased f	rom	Lease start date	Lease end date
					m/d/y	m/d/y
	~ .			-		
Declaration	n - Corporation	<u>n</u>		Dec	laration - Non-Resident Pe	rson
I hereby de	clare that the	information	1	I he	reby declare that the inform	nation provided above
	bove and in an				in any attachment is accura	
accurate and true to the best of my		my	knowledge and belief for the	he reporting calendar		
knowledge and belief for the reporting		year	·			
calendar year						
Signature on behalf of the Lessee			Sign	nature of Disclosing Lessee	*	
Corporation *						
Date		Date				
Date			Daic			
*This form must be signed by an authorized		*This form must be signed by the Disclosing Lessee*				
signing officer of the Lessee Corporation.						
		<u></u>				
Name of Signatory (Please print)			Nan	ne (Please print)		
Title						
Title						

70

Cap. L-5